Author's response to reviews

Title: The impact of global health initiatives on trust in health care provision under extreme resource scarcity: presenting an agenda for debate from a case study of emergency obstetric care in Northern Tanzania

Authors:

Øystein E Olsen (oystein.olsen@cih.uib.no)

Version: 8 Date: 9 May 2010

Author's response to reviews: see over
The Editors
Health Research Policy and Systems
BioMed Central

RE: Submission of revised version of the article “Challenging common thinking on health care provision under extreme resource scarcity: a case study of emergency obstetric care in Northern Tanzania”

Dear Sir / Madam

I hereby submit to you for review again the revised article

"The impact of Global Health Initiatives on trust in health care provision under extreme resource scarcity: presenting an agenda for debate from a case study of emergency obstetric care in Northern Tanzania"

The paper has been revised in line with the last review received upon submission of the second draft, asking only for the paper to be edited by a native English speaker. I have now done this, and am pleased to submit to you again the paper with subsequently improved language.

Please find enclosed the revised article and tables and figures in electronic form.

Yours Sincerely

(sign)
Dr. Øystein Evjen Olsen
Research Fellow, MD, MSc, HPPF (DLSHTM)
Center for International Health
University of Bergen
N-5021 Norway

oystein.olsen@cih.uib.no